

Overburdened judges

Ludivine Roussey, Raphaël Soubeyran

▶ To cite this version:

Ludivine Roussey, Raphaël Soubeyran. Overburdened judges. 2018. halshs-01936006

HAL Id: halshs-01936006 https://shs.hal.science/halshs-01936006

Preprint submitted on 27 Nov 2018

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Overburdened judges

Ludivine Roussey & Raphael Soubeyran



CEE-M Working Paper 2018-03









Overburdened Judges*

Ludivine Roussey[†] Raphael Soubeyran[‡]

This version: October 2017

Abstract. We develop a double-sided moral hazard model in which the production of justice depends on two tasks (jurisdictional and administrative). The jurisdictional task can be provided only by a judge (the agent) while the administrative task can be provided either by the government (the principal) and/or by the judge. However, the judge performs the administrative task at a higher unit cost. First, we show that the first-best situation is such that the judge exerts no effort to provide the administrative task. Second, we show that two forms of (second-best) optimal contract can emerge when neither the government's effort nor the judge's effort is contractible: either the incentives are shared between the government and the judge and the judge exerts no effort to provide the administrative task, or the judge faces high-powered incentives which induce her to exert effort to provide both tasks. Our model proposes a rationale for judges work overload observed in many countries.

JEL classification: double-sided moral hazard, task misallocation, judicial organization, production of judicial services.

Keywords: D20, D86, K40.

^{*}We thank Eshien Chong, Marie-Cecile Fagart, Etienne Lehmann, and Antoine Soubeyran, and participants at various seminars and conferences, for very helpful discussions and suggestions. We thank Wale Arowolo for his careful reading of the article. Finally, we thank two anonymous referees for their constructive comments and suggestions.

[†]LIRAES, University Paris Descartes, Sorbonne Paris Cité University. E-mail: ludivine.roussey@parisdescartes.fr.

[‡]CEE-M, Univ. Montpellier, CNRS, INRA, Montpellier SupAgro. E-mail: raphael.soubeyran@inra.fr.

1 Introduction

"In an ideal world, a trial would never be unreasonably delayed or cut short. Judges would never need to juggle multiple difficult trials or drown in administrative tasks that distract from the fair adjudication of cases" Tomi Mendel (2015, Vanderbilt Law Review)

In reality, judges are frequently diverted from their judicial function (Buscaglia and Ulen, 1997; Smith and Feldman, 2001; French, 2009). They are often required to spend time performing administrative tasks in addition to their jurisdictional tasks. Consequently, they may become overburdened (Sheldon, 1981). Since this causes backlogs and delays, the judges' administrative workloads can raise the cost of access to the judiciary, and reduce faith in the legal system (Dakolias, 1999). In turn, these inefficiencies in the judiciary are likely to have negative impacts on investments, financial markets and economic growth (North, 1990; La Porta et al., 1997, 1998; Djankov et al., 2003).

One of the reasons why such a situation has arisen is lack of public resources and court administrative staff to carry out non-judicial duties. Smith and Feldman (2001) report that state supreme courts in the U.S. lack equivalent administrative assistance to that provided to the federal judiciary, and are thus more likely to bear the administrative burden. Furthermore, the Strategic Plan for the Federal Judiciary of the United States³ refers to the importance of securing sufficient resources to attract, recruit and retain the necessary staff to support judges and handle the workload. Similarly, a report from the French Senate⁴ interprets the small numbers of law clerks per judge as resulting in lack of logistic support, and points to the potentially disastrous consequences for judges' workloads and the functioning of the whole justice system.

There is evidence of a negative relationship between judges' administrative workload and judiciary performance. Buscaglia and Dakolias (1999) show that clearance rates and procedural times improve if judges spend less time on administrative tasks and focus more on their adjudicative duties. Flanders (1977) shows that the number of completed cases per judge decreases with the number of published decisions and concludes that "many judges may wish to consider reducing the number of opinions they prepare, in the interest of conserving their time to meet the demands of other cases on their dockets". Gomes and Akutsu (2016) argue that an increase in the number of assistants increases the time that judges can devote to judicial activities, and for the case of Brazil show a positive link between the number of assistants per judge and courts' productivity.

This situation raises a number of questions. First, why are judges ultimately diverted from their judicial function? Second, given that judges' opportunity cost of performing administrative tasks is clearly high, why do governments not provide more support staff to carry out these administrative duties? Finally, why do judges agree to take on these duties in addition to their judicial work? In the present paper, we provide a rationale for this puzzle.

We develop a double-sided moral hazard model in which the production of justice depends on two tasks. One of these tasks, the jurisdictional one, can be provided only by a judge (the agent).

¹Buscaglia and Ulen (1997) report that in some Latin American countries such as Argentina, Brazil and Peru, judges spend more than 60% of their working time on non-adjudicative tasks.

²In the long run, the burden imposed on judges is also likely to decrease the incentives for legal professionals to work in the public legal system, and consequently, to lower the quality of human resources in the judiciary.

³http://www.uscourts.gov/statistics-reports/strategic-plan-federal-judiciary

⁴http://www.senat.fr/rap/a10-116-4/a10-116-412.html

The other task, the administrative one, can be provided by the government (the principal) and/or by the judge but at a higher unit cost. First, we show that the first-best situation is that the judge does not exert any efforts to provide the administrative task.

Thereafter, we show that two forms of optimal contract can emerge when the effort levels of the government and of the judge are not contractible: either the judge faces a high incentive (i.e. the judge is completely responsible for the judicial output) which induces her to exert effort to provide the two tasks, or the incentives are shared between the judge and the government (i.e. they share responsibility for the judicial output) and the judge exerts no efforts to provide the administrative task.

We show that these two types of contract are inefficient for two different reasons. The former is inefficient because there is a misallocation of tasks compared to the first-best situation where the judge does not perform any administrative tasks. The latter is inefficient because - although the allocation of tasks is the same as in the first-best situation - neither the government nor the judge receives full return for their effort. Consequently, neither party provides the first-best level of effort. When choosing the optimal contract, the government faces a trade-off between these two inefficiencies. We show that the optimal contract (i.e. the contract preferred by the government among these two types of contract) depends on the difference between the government's and the judge's unit cost for performing the administrative task. If the differential is large, the optimal contract is the one in which the incentives are shared and the judge does not perform any administrative tasks. If the differential is small, the optimal contract is the one in which the judge faces high-powered incentives and performs both the jurisdictional and administrative tasks.

Using a Cobb-Douglas specification for the judicial production, we provide some additional results to show the link between the form of the optimal contract and the fundamentals of the model, i.e. the output elasticities of the jurisdictional and administrative tasks, and the unit cost parameters. This specification allows us to provide some insights into the relationship between the level of the incentives faced by the judge, the degree of inefficiency associated with each type of optimal contract, and the fundamentals of the model. Specifically, we find that the inefficiency due to the misallocation of administrative tasks increases when the cost of the administrative tasks for the government, and that the degree of inefficiency due to the (double) moral hazard problem is larger when the judicial output elasticities of the administrative and jurisdictional effort are high.

The rest of the paper is organized as follows. Section 2 reviews the related literature. Our contracting model of judicial production is described in Section 3. Section 4 presents our main results, and Section 6 discusses possible remedies for the problem at hand, and proposes several testable implications of our analysis. Section 5 proposes two extensions of our model and Section 7 concludes.

2 Related literature

Our paper contributes to several strands of the law and economics literature and also to the literature on organization theory.

It adds to work that focuses on the role of the organization of courts and its impact on the efficiency of the judicial sector. This literature explores the link between various performance

measures of judicial output and aspects of judicial organization such as the appointment process (Beenstock and Haitovsky, 2004), career incentives (Schneider, 2005), case-management (Buscaglia and Ulen, 1997), mode of disposition (Dimitrova-Grajzl et al., 2014), court size (Peyrache and Zago, 2016; Dimitrova-Grajzl et al., 2012; Grajzl and Silwal, 2016), and resources allocated to courts (Deffains and Roussey, 2011; Roussey and Deffains, 2012). We contribute to this literature by studying how incentives and the resulting allocation of administrative tasks between a representative judge and the government affect judicial efficiency in a setting with moral hazard on both sides.

It is noteworthy that the judicial output should be considered in our model as a composite variable including the different aspects of justice. Research studying the measurement of judicial output refers to the need to consider its different aspects (Dakolias, 1999; Roussey, 2016). Basically, three types of judicial output index can be distinguished. First, judicial output can be considered in terms of dispute resolution which can be measured by the average time needed to resolve a case (Mitsopoulos and Pelagidis, 2007), or the number of cases a judge completes in a given period (Beenstock and Haitovsky, 2004; Rosales-López, 2008; Dimitrova-Grajzl et al., 2012). Second, judicial output can be measured by the quality of decisions. This can be captured by the percentage of cases confirmed on appeal (Schneider, 2005), or the occurrence of legal errors (Chopard et al., 2018). Third, judicial output can be measured in terms of lawmaking (Landes and Posner, 1979). Even in civil-law countries, it is now widely acknowledged that judges participate in the creation of law by interpreting statute law, and by publishing new judicial opinions. This output can be measured by the number of decisions published by a judge or the number of citations to a decision (Choi et al., 2010).

Our approach is also in line with the literature on how incentives influence a judge's activity and behavior (Dewatripont and Tirole, 1999; Shapiro and Levy, 1995). This literature strand highlights the role of financial rewards (Cooter, 1983) but also prestige, individual and collective reputation (Garoupa and Ginsburg, 2010), popularity (Posner, 1993) and career opportunities (Taha, 2004; Schneider, 2005) as motivational factors. To the extent that judges' statuses differ across legal systems, rules likely to motivate judges differ between common law and civil law countries. In Spain, where the civil-law judiciary employs tenured judges on fixed pay, Bagues and Esteve-Volart (2010) show that between 2004 and 2007 financial bonuses awarded to judges achieving good quantitative targets increased average judicial production. Posner (1993) suggests that since in common law countries reputation and prestige can be effective motivations for elected judges, greater transparency in the conduct of their decision-making could provide very strong incentives. Relying on evidence from the New York State Committee to Review Audio-Visual Coverage of Court Proceedings (1997) that allowing trials to be broadcasted is linked positively to better judicial performance, Burki and Perry (1998) argue that judges exposed to press and public scrutiny have an incentive to do a better job in conducting trial and proceedings. They argue also that disclosing how individual judges vote if decisions are made by a panel of judges could incentivize judges to consider and decide cases more carefully. Schneider (2005) describes the link between the promotion system within the German labor judiciary and appraisals of judges' quantitative performance. In particular, the odds of promotion for judges acting in the Labor Courts of Appeals are based on individual quantitative productivity data such as number of cases processed, backlogs and settlement rates. Taha (2004) presents empirical evidence from U.S.

federal district courts that judges' performance is influenced by their chances of promotion to a U.S. courts of appeals. The present paper builds on this literature and considers that government can choose the incentives offered to judges. We show that, depending on the relative cost of administrative tasks for the government, the cost of jurisdictional tasks, and the form of the judicial production function, the optimal incentives the judge should face are either high-powered or low-powered. In particular, if the judge cost of performing administrative tasks is sufficiently low (but still higher than the cost of government performing administrative tasks), then the incentives should be high. Otherwise, the incentives should be low.

Finally, this paper contributes to the double-sided moral hazard literature which considers situations where both a principal and an agent provide efforts in order to produce an output but their effort levels are not verifiable. This setting differs from the standard principal-agent model in which only the agent provides an effort. So far, double-sided moral hazard models have helped to explain the prevalence of sharecropping in agriculture (Stiglitz, 1974; Eswaran and Kotwal, 1985), the emergence of franchise contracts (Mathewson and Winter, 1985; Lal, 1990; Bhattacharyya and Lafontaine, 1995) and the (in)efficiency of teams and partnerships (Holmstrom, 1982; Legros and Matsushima, 1991; Legros and Matthews, 1993). In the present paper, we develop a double-sided moral hazard model to study the organization of the judiciary. Our model departs from the usual double-sided moral hazard model because we assume that both the principal and the agent can perform one of the tasks. This leads to the novel conclusion that, either the principal chooses to implement high-powered incentives to induce the agent to perform both tasks, or chooses to share the incentives in order to induce the agent to perform the task that the principal is not capable of doing, while the principal performs the other task. If the principal chooses to induce the agent to perform both tasks, the optimal contract is inefficient, as in the usual double-sided moral hazard problem. However, in this case, unlike in the double-sided moral hazard problem, the inefficiency is due to task misallocation.

3 The Model

Judicial services are provided by the judiciary by combining two kinds of tasks, administrative tasks and jurisdictional tasks, which are performed thanks to efforts A and e, respectively. These variables can include both quantity effort and quality effort. The two types of task require different types of competencies. Jurisdictional tasks e can be performed only by a skilled representative judge (J) and consist of preparing case files, hearing parties, and rendering and writing judgments. Administrative tasks can be performed either by the judge or by administrative staff such as clerks, judicial assistants and secretaries responsible for the preparation, registration or filing of cases. We assume that the government (G) can choose the number of administrative tasks performed by the administrative staff (on the basis that each administrative agent performs a fixed number of administrative tasks and that G chooses the number of administrative agents).⁵ The number of administrative tasks performed by J is denoted by $a \ge 0$ and the number of administrative tasks performed by J is denoted by J is denoted by J is necessary of the number of administrative agents an exogenous minimum level of administrative effort that J is obliged to perform. It can represent the level of administrative effort that the government pays ex-ante (before setting the contract), or the level of administrative

⁵We abstract from the principal-agent relationship between the government and the administrative staff.

effort that the government commits to before the judge's incentive scheme is chosen. In Section 6 we assume that this level is exogenous and discuss how it could be endogenously chosen. We assume that the administrative efforts provided by J and G are perfect substitutes:

$$A = a + r + \underline{r}. (1)$$

We model the need for both jurisdictional and administrative efforts to provide judicial services using a production function f(A, e) which increases when the total administrative effort increases and when the jurisdictional effort increases, $f_A, f_e > 0$ (with subscripts denoting partial derivatives). We assume that f is strictly concave with respect to (A, e). We assume also that $f_e(A, 0) > c'(0)$ for all $A \ge 0$ in order to focus on solutions such that e > 0. Notice that for most of our results, we do not impose any more structure on the production function. In particular, we make no assumption regarding the sign of f_{Ae} .

The benefit from the judicial production is

$$\pi(A, e) = f(A, e) + \tilde{\epsilon}, \tag{2}$$

where $\tilde{\epsilon}$ is a random term with mean zero.

Note that f can be interpreted as a measure of judicial output in terms of productivity, quality and/or judicial lawmaking (see Section 2 for some related literature).

The expected benefit from the production of justice then is

$$E[\pi] = f(A, e). \tag{3}$$

We suppose that $\tilde{\epsilon}$ cannot be observed by G or J. Moreover, the effort level exerted by one co-provider is not verifiable. Because the effort levels and the random term are non-verifiable, observing the output f does not allow one to infer the effort level exerted by G and J hence the incentive problem. Moreover, as a consequence, the incentive contract must be based on f and not on efforts a, e or r.

We assume that J can allocate her time between the two tasks (administrative and jurisdictional). We assume also that J faces no time constraint. In Section 5.1 we will show that including a time constraint would not modify our qualitative results.

The unit cost for J to perform an administrative task is denoted $\delta_a > 0$, and the cost of performing a jurisdictional task is denoted c(e), with c' > 0 and $c'' \ge 0$. The unit cost for G to perform an administrative task is $\delta_r > 0$. Since performing non-judicial duties such as administrative tasks is not the main function of judges, their opportunity cost is quite large. Thus, we assume that the unit cost for J to perform administrative tasks is always larger than the unit cost for G to perform administrative tasks, i.e. $\delta_r < \delta_a$.

J receives a fixed salary F, which may be supplemented by a variable remuneration βf depending on the output of the judiciary. The variable remuneration can be financial or in the form of psychological satisfaction, reputation enhancement or promotion chances, depending on the performance of the judiciary (see Section 2 for a review of the literature on this point). The rate β of the flexible remuneration is decided by G (who also pays F), and together with F defines the incentive linear contract.⁶ G values the judicial output f(A, e), pays a remuneration to J,

⁶This assumption does not imply loss of generality. We build on Bhattacharyya and Lafontaine (1995) who

 $\beta f(A, e) + F$ and carries the cost of its administrative effort, $\delta_r r$. The value $(1 - \beta) f$ represents the benefit that G can expect from the judicial output. It can be regarded as the merit – or accountability – that citizens attribute to G for the performance of the judiciary. This ultimately might result in a certain degree of popularity of G, or improve the likelihood of re-election.

The timing of the game is as follows. First G decides on the incentive contract. Then, J decides whether or not to accept the contract. Finally, if the contract is accepted by J, both providers G and J choose their preferred effort levels given the agreed contract: J chooses e and a and G chooses e. Then, the value of the judicial output is observed and G and G are their respective payoffs.

When it designs the contract, G anticipates this move. Thus, we can write G's problem as:⁷

$$Max_{\beta,F,r>0,a>0,e>0} Eu^G = (1-\beta)f(A,e) - \delta_r r - F,$$
(P)

subject to:

$$Eu^{J} = F + \beta f(A, e) - \delta_{a}a - c(e) \ge 0, \tag{PC}$$

$$a\frac{\partial Eu^J}{\partial a} = 0 = e\frac{\partial Eu^J}{\partial e},\tag{IC_J}$$

and,

$$r\frac{\partial Eu^G}{\partial r} = 0, (IC_G)$$

Condition (PC) is J's participation constraint. It states that J will only accept a contract that provides her with a positive expected utility. (IC_J) is J's incentive constraint which determines her effort. Condition (IC_G) is G's incentive constraint. The incentive constraints also include three conditions that we do not include in the program, namely $\frac{\partial Eu^J}{\partial a} \leq 0$, $\frac{\partial Eu^J}{\partial e} \leq 0$, and $\frac{\partial Eu^G}{\partial r} \leq 0$. These conditions have to be checked when the corresponding effort levels are null.

It can be seen that necessarily, at the optimum, the participation constraint is binding, i.e. (PC) becomes

$$Eu^{J} = F + \beta f(A, e) - \delta_a a - c(e) = 0.$$

$$\tag{4}$$

As a consequence, G's optimization program is equivalent to maximizing the joint surplus, $S \equiv Eu^G + Eu^J$ with respect to (β, r, a, e) , subject to the incentive constraints (IC_J) and (IC_G) . Condition (4) characterizes the lump-sum transfer F.

The Lagrangian of this simplified problem is given by:

$$L = f(A, e) - \delta_a a - c(e) - \delta_r r + \lambda_a a \frac{\partial E u^J}{\partial a} + \lambda_e e \frac{\partial E u^J}{\partial e} + \lambda_r r \frac{\partial E u^G}{\partial r} + \mu_r r + \mu_a a + \mu_e e, \quad (5)$$

where λ_a and λ_e are the Lagrange multipliers associated to J's incentive constraint (IC_J) , λ_r is the Lagrange multiplier associated to G's incentive constraint (IC_G) , and μ_r, μ_a , and μ_e are the Lagrange multipliers associated to the positivity constraints respectively $r \geq 0$, $a \geq 0$ and $e \geq 0$,.

The necessary conditions include the following.

show that in double-sided moral hazard contracting problems there is always an optimal contract that is linear. 7 We drop the arguments of f when there is no possible confusion.

With respect to J's administrative effort a:

$$\frac{\partial L}{\partial a} = f_A - \delta_a + \lambda_a \left(\beta f_A - \delta_a + a\beta f_{AA}\right) + \lambda_e e\beta f_{Ae} + \lambda_r r(1 - \beta) f_{AA} + \mu_a = 0. \tag{NCa}$$

With respect to J's jurisdictional effort e:

$$\frac{\partial L}{\partial e} = f_e - c' + \lambda_a a \beta f_{Ae} + \lambda_e \left(\beta f_e - c' + \beta e f_{ee} - e c'' \right) + \lambda_r r (1 - \beta) f_{Ae} + \mu_e = 0. \tag{NC_e}$$

With respect to G's administrative effort r:

$$\frac{\partial L}{\partial r} = f_A - \delta_r + \lambda_a a \beta f_{AA} + \lambda_e e \beta f_{Ae} + \lambda_r \left[(1 - \beta) f_A - \delta_r + r(1 - \beta) f_{AA} \right] + \mu_r = 0. \quad (NC_r)$$

With respect to β :

$$\frac{\partial L}{\partial \beta} = \lambda_a a f_A + \lambda_e e f_e - \lambda_r r f_A = 0. \tag{NC}_{\beta}$$

The necessary conditions associated with the positivity constraints are $\mu_a a = 0$, $\mu_e e = 0$, $\mu_r r = 0$ and $\mu_a, \mu_e, \mu_r, a, e, r \ge 0$.

The solution to G's optimization problem is not straightforward because the Lagrangian is not necessarily concave. Therefore, as we show in Section 4, the optimal contract can induce J either to specialize in jurisdictional tasks or to perform both kinds of tasks.

4 Results

In this section, we proceed with the analysis. We focus first on the first-best situation where the efforts of G and J are verifiable, then we investigate the situation where the efforts of G and J are not contractible.

4.1 First Best

We consider the first-best situation in which effort levels are verifiable, and thus can be included in the contract. Let a^{FB} , e^{FB} and r^{FB} denote the first-best levels of effort. They maximize G's and J's joint surplus:

$$\max_{r \ge 0, a \ge 0, e \ge 0} S = f(A, e) - \delta_a a - c(e) - \delta_r r, \tag{FB}$$

where $A = a + r + \underline{r}$.

We obtain the following preliminary result:

Proposition 1 [First best]: The first-best solution is unique and such that the judge does not perform any administrative tasks, $a^{FB} = 0$.

Proof: Since the administrative efforts of G and the judge are perfect substitutes $(A = a + r + \underline{r})$, and the marginal costs are constant with $\delta_a > \delta_r$, we must have $a^{FB} = 0$.

This sharp result holds because the administrative efforts of G and J are perfect substitutes, and the cost of performing administrative tasks is linear and strictly smaller for G. We relax these assumptions in Section 5.

In order to focus on the most interesting situation, we make the following assumption:

Assumption 1: The minimum level of administrative effort provided by the government, \underline{r} , is such that, $f_A(\underline{r}, e) > \delta_a$ for all $e \ge 0$.

Assumption 1 states that, when the administrative total effort is minimal $(A = \underline{r})$, the marginal judicial benefit of an increase in the administrative effort is always larger than the unit cost for J performing administrative tasks. This implies that the optimal total administrative effort is always interior, i.e. $A^* > r$.

The first-best effort levels are characterized by the following conditions:

$$a^{FB} = 0, \ f_A(r^{FB} + \underline{r}, e^{FB}) = \delta_r \text{ and } f_e(r^{FB} + \underline{r}, e^{FB}) = c'(e^{FB}).$$
 (6)

In the first-best situation, J performs only jurisdictional tasks and the levels of the administrative and jurisdictional efforts are such that the marginal judicial benefit of an increase in each kind of effort equals its marginal cost.

We next analyze the optimal contract when the three effort levels, a, r and e, are not contractible. Notice that none of our main results requires either the production function f or the jurisdictional effort cost function e to be specified.

4.2 Optimal contract: a trade-off between two inefficiencies

Here we focus on the situation where the efforts of G and of J are not contractible. Using the necessary conditions provided in Section 3, we obtain the following result:

Proposition 2 [Allocation of tasks]: The optimal contract can lead to two different task allocations: the judge either performs both administrative and jurisdictional tasks, or only jurisdictional tasks. Specifically, the optimal contract is such that either (C1) the judge performs the two tasks and government's administrative effort is minimal ($a^* > 0$ $e^* > 0$ and $r^* = 0$) and the judge is a full-residual claimant ($\beta^* = 1$), or (C2) the judge performs only jurisdictional tasks ($a^* = 0$ $e^* > 0$ and $r^* > 0$) and the incentives are shared ($0 < \beta^* < 1$).

Proposition 2 states that two types of contract can be optimal. The first type of contract (C1) is such that J faces high-powered incentives ($\beta^* = 1$) and then is motivated to perform both jurisdictional and administrative tasks. The optimal effort levels are such that

$$r^* = 0$$
, $f_A(a^* + \underline{r}, e^*) = \delta_a$ and $f_e(a^* + \underline{r}, e^*) = c'(e^*)$. (7)

These conditions state that G performs the minimum administrative tasks (\underline{r}) , and J performs administrative tasks $(a^* > 0)$ and jurisdictional tasks. The levels of administrative and jurisdictional efforts are such that the marginal judicial benefit of an increase in each kind of effort equals its marginal cost. These differ from the first-best conditions (6) because the marginal cost of the administrative effort is δ_a instead of δ_r . So, J faces high-powered incentives and then performs the administrative tasks but G performs only the minimum administrative tasks r.

⁸This assumption also implies that $\underline{r} < A^{FB}$. In Section 6, we discuss the case where $\underline{r} = A^{FB}$.

The second type of contract (C2) is such that incentives are shared between J and G (0 < β^* < 1) and the optimal effort levels are characterized by:

$$a^* = 0, (1 - \beta)(f_A(r^* + \underline{r}, e^*)) = \delta_r \text{ and } \beta f_e(r^* + \underline{r}, e^*) = c'(e^*).$$
 (8)

These conditions state that J does not perform any administrative tasks, G performs a higher level of administrative tasks than the minimum $(r^* > 0)$, and J performs jurisdictional tasks. The levels of the administrative and jurisdictional efforts are such that the weighted marginal judicial benefit of an increase in each kind of effort equals its marginal cost. They differ from the first-best conditions (6) because the marginal benefits are distorted due to the shared incentives.

Notice that Proposition 2 also implies that above the minimum administrative effort, \underline{r} the remaining administrative effort cannot be shared between J and G. This result is due to the combination of perfect substitutability between J's and G's administrative efforts and the linearity of the administrative effort costs. We relax these assumptions in Section 5.

We focus now on the difference between the optimal effort levels and the first-best effort levels. We obtain the following result:

Corollary 1 [Optimal efforts]: If $f_{eA} \ge 0$, then both the optimal administrative effort and the optimal jurisdictional effort are lower than the corresponding first-best levels of effort, i.e. $A^* < A^{FB}$ and $e^* \le e^{FB}$. We have $e^* = e^{FB}$ only if $f_{eA} = 0$ under contract (C1).

This result states that the optimal efforts are lower than the first-best levels when the judicial production function is such that a marginal increase in the administrative effort level does not decrease the productivity of jurisdictional effort, and vice versa ($f_{eA} \ge 0$). The intuition is as follows.

First, consider the case of contract (C1). In this situation, J exerts effort on both tasks. Since the unit cost for J to perform administrative tasks is larger than the unit cost for G to perform these tasks, J exerts a lower administrative effort level than the effort level that G provides in the first-best situation. This does not influence the choice of the jurisdictional effort level as long as $f_{eA} = 0$. However, if $f_{eA} > 0$, the decrease in the administrative effort decreases the marginal benefit of jurisdictional efforts, and then J chooses a jurisdictional effort level below the first-best level.

Second, consider the case of contract (C2). In this situation, J specializes in jurisdictional tasks while G performs administrative tasks. Thus, the allocation of tasks corresponds to the first best situation but J and G receive only a fraction of the return from their efforts, and then they exert lower levels of effort than in the first best situation. If $f_{eA} > 0$, the decrease in the levels of each type of effort decreases the marginal benefit of an increase in the level of the other type of effort, which then magnifies the decrease in the administrative and jurisdictional efforts.

In the case (not considered here) where the judicial production function is such that a marginal increase in the administrative effort level decreases the productivity of the jurisdictional efforts and vice versa (i.e. $f_{eA} < 0$), we cannot exclude the possibility that either the optimal administrative effort level or the optimal jurisdictional effort level will be larger than the corresponding first-best level of effort. The reason for this is that, in this situation, a decrease in the administrative effort level provides incentives to increase the jurisdictional level and vice versa.

We know from Proposition 2 that either type of contract (C1 or C2) could emerge as the optimal contract. The following result shows that the optimal contract is generically unique and that the type of contract that emerges as optimal depends on a specific threshold:

Proposition 3 [Optimal contract]: There exists a threshold $\hat{\theta} > \delta_r$ such that the optimal contract is the non-specialization contract (C1) if $\delta_a < \hat{\theta}$ and the optimal contract is the specialization contract (C2) if $\delta_a > \hat{\theta}$. If $\hat{\theta} = \delta_a$, the government is indifferent between contract (C1) and contract (C2).

Proposition 3 states that the non-specialization contract with high-powered incentives for J ($\beta^* = 1$) is chosen by G if the unit cost of the administrative effort for J is sufficiently low (but still above the unit cost of the administrative effort for G).

This arises because G faces a trade-off between the inefficiency due to misallocation of tasks, and the inefficiency due to moral hazard. If the unit cost of the judge's administrative effort is sufficiently low (but still above the unit cost of G's administrative effort), then the loss of surplus due to the misallocation of tasks is lower than the loss of surplus due to the (double-sided) moral hazard problem, and as a consequence, G chooses the non-specialization contract (C1). If the unit cost of the judge's administrative effort is sufficiently large, then the loss of surplus due to task misallocation is larger than the loss of surplus due to the (double-sided) moral hazard problem, and therefore, G chooses the specialization contract (C2).

The threshold $\widehat{\theta}$ is the level of J's unit cost of administrative effort such that the joint surplus is the same under either contract (C1) or contract (C2), or:

$$S\left(A_1^*\left(\widehat{\theta}\right), e_1^*\left(\widehat{\theta}\right)\right) = S\left(A_2^*, e_2^*\right),\tag{9}$$

where $A_1^* = a_1^* + \underline{r}$ and $A_2^* = r_2^* + \underline{r}$. (A_1^*, e_1^*) is characterized by $f_A(A_1^*, e_1^*) = \widehat{\theta}$ and $f_e(A_1^*, e_1^*) = c'(e_1^*)$. (A_2^*, e_2^*) is characterized by $(1 - \beta^*)f_A(A_2^*, e_2^*) = \delta_r$ and $\beta^*f_e(A_2^*, e_2^*) = c'(e_2^*)$, with the optimal level of incentives associated with contract (C2), β^* , characterized as follows. If $(f_A)^2f_{ee} - (f_e)^2f_{AA} = 0$ then:

$$\beta^* = \frac{-(f_e)^2 f_{AA}}{(f_A)^2 c'' - 2(f_e)^2 f_{AA}},\tag{10}$$

with $(A, e) = (A_2^*, e_2^*)$.

Otherwise, we have:

$$\beta^* = \frac{(f_A)^2 c'' - 2(f_e)^2 f_{AA} - f_A \sqrt{(f_A)^2 (c'')^2 - 4(f_e)^2 f_{AA} (c'' - f_{ee})}}{2 \left[(f_A)^2 f_{ee} - (f_e)^2 f_{AA} \right]},\tag{11}$$

with $(A, e) = (A_2^*, e_2^*)$.

Thus, the threshold $\widehat{\theta}$ depends in a non-trivial way, on all the fundamentals of the model: the judicial production function, the administrative effort unit cost parameter δ_r and the jurisdictional effort cost function c(.).

While the above analysis provides some general insights, it fails to provide the intuition behind all the factors determining the optimal contract without further structure on the judicial production function and the jurisdictional effort cost function. The general formulation given above does not provide closed-form expressions for the levels of the optimal efforts, the optimal incentives level, the threshold that characterizes the optimal contract or the total surplus. To obtain further intuition, we analyze the solution to our problem using specific functional forms.

4.3 Cobb-Douglas judicial output function and linear cost of jurisdictional effort

In this section, we use a Cobb-Douglas specification of the production function f and a linear specification of the jurisdictional effort cost function c(.) in order to illustrate our general results and to provide some comparative statics.

Let us assume that the production function is such that $f(A, e) = A^{\alpha}e^{\gamma}$ with $\alpha + \gamma < 1$ and $c(e) = \delta_e e$ with $\delta_e > 0$. Using this example, we find that the first-best levels of effort are given by:

$$r^{FB} = \left((\gamma/\delta_e)^{\gamma} (\alpha/\delta_r)^{1-\gamma} \right)^{\frac{1}{1-(\alpha+\gamma)}} - \underline{r} \text{ and } e^{FB} = \left((\alpha/\delta_r)^{\alpha} (\gamma/\delta_e)^{1-\alpha} \right)^{\frac{1}{1-(\alpha+\gamma)}}. \tag{12}$$

Notice that here, Assumption 1 implies that $\underline{r} < ((\gamma/\delta_e)^{\gamma}(\alpha/\delta_a)^{1-\gamma})^{\frac{1}{1-(\alpha+\gamma)}}$. Moreover, in the first best case, the total surplus is given by:

$$S^{FB} = \left[1 - (\alpha + \gamma)\right] \left[\left(\frac{\gamma}{\delta_e}\right)^{\gamma} \left(\frac{\alpha}{\delta_r}\right)^{\alpha} \right]^{\frac{1}{1 - (\alpha + \gamma)}}.$$
 (13)

We find that the optimal effort levels under contract (C1) are given by:

$$a_1^* = \left((\gamma/\delta_e)^{\gamma} (\alpha/\delta_a)^{1-\gamma} \right)^{\frac{1}{1-(\alpha+\gamma)}} - \underline{r} \text{ and } e_1^* = \left((\alpha/\delta_a)^{\alpha} (\gamma/\delta_e)^{1-\alpha} \right)^{\frac{1}{1-(\alpha+\gamma)}}. \tag{14}$$

These expressions are close to the expressions of the first-best level of G's administrative effort (r^{FB}) and the first best-level of J's jurisdictional effort (e^{FB}) , respectively. However, the unit cost of administrative effort differs. Indeed, J performs administrative task and her unit cost of administrative effort is larger than G's $(\delta_a > \delta_r)$. As a consequence, the level of administrative effort is lower than the corresponding first-best level. Since $f_{eA} > 0$ when f takes the Cobb-Douglas form, the level of jurisdictional effort is also lower than the corresponding first best level. This illustrates the task misallocation problem that arises under contract (C1).

Now, let us consider contract (C2). In this case, we find that the optimal levels of effort are given by:

$$r_2^* = \left((1 - \beta_2^*) (\gamma/\delta_e)^{\gamma} (\alpha/\delta_r)^{1-\gamma} \right)^{\frac{1}{1-(\alpha+\gamma)}} - \underline{r} \text{ and } e_2^* = \left(\beta_2^* (\alpha/\delta_r)^{\alpha} (\gamma/\delta_e)^{1-\alpha} \right)^{\frac{1}{1-(\alpha+\gamma)}}. \tag{15}$$

where $0 < \beta_2^* < 1$. Notice that both levels of effort are lower compared to the first-best levels only because the incentives are shared between J and G. This illustrates the (double-sided) moral hazard inefficiency that arises in contract (C2).

We find that the optimal level of incentives β_2^* is given by:

$$\beta_2^* = 1/\left(1 + \sqrt{\frac{\alpha}{\gamma} \frac{1-\gamma}{1-\alpha}}\right). \tag{16}$$

Notice that, if $\alpha = \gamma$, incentives are shared equally between J and G, $\beta_2^* = 1/2$. Moreover, J's incentives level, β_2^* , decreases with the output elasticity of the administrative effort, α , and increases with the output elasticity of jurisdictional effort, γ . Consequently, G's incentives level,

 $(1 - \beta_2^*)$, increases with the output elasticity of administrative effort, α , and decreases with the output elasticity of jurisdictional effort, γ . This result is quite intuitive: the incentive sharing rule should favor the (best) provider of the task with the largest output elasticity of effort.

Now let us consider the total surplus in each case. In the case of contract (C1), the total surplus is given by:

$$S_1^* = \left[1 - (\alpha + \gamma)\right] \left[\left(\frac{\gamma}{\delta_e}\right)^{\gamma} \left(\frac{\alpha}{\delta_a}\right)^{\alpha} \right]^{\frac{1}{1 - (\alpha + \gamma)}},\tag{17}$$

and, in the case of contract (C2), the total surplus is:

$$S_{2}^{*} = \left[\left[\left[\left(1 - \beta_{2}^{*} \right)^{\alpha} \left(\beta_{2}^{*} \right)^{\gamma} \right]^{\frac{1}{1 - (\alpha + \gamma)}} - \alpha \left(1 - \beta_{2}^{*} \right)^{\frac{1}{1 - (\alpha + \gamma)}} - \gamma \left(\beta_{2}^{*} \right)^{\frac{1}{1 - (\alpha + \gamma)}} \right] \left[\left(\frac{\gamma}{\delta_{e}} \right)^{\gamma} \left(\frac{\alpha}{\delta_{r}} \right)^{\alpha} \right]^{\frac{1}{1 - (\alpha + \gamma)}}. \tag{18}$$

Having computed the closed-form expressions of the efforts and total surpluses in both the first-best case and for the two possible optimal contracts, we can provide further intuition about how the threshold level $\hat{\theta}$ depends on the parameters of the model and about the extent of the two types of inefficiency.

The threshold $\widehat{\theta}$ is a non-monotonic function of α and γ : In order to provide additional insights in relation to the result of Proposition 3, we compute the closed form expression of the threshold $\widehat{\theta}$ and analyze its dependence on the fundamentals of the model.

Contract (C1) is chosen if and only if $S_1^* \geq S_2^*$, or $\delta_a \leq \widehat{\theta}$ with:

$$\widehat{\theta} = \delta_r \left[\frac{1 - (\alpha + \gamma)}{\left[(1 - \beta_2^*)^{\alpha} (\beta_2^*)^{\gamma} \right]^{\frac{1}{1 - (\alpha + \gamma)}} - \alpha (1 - \beta_2^*)^{\frac{1}{1 - (\alpha + \gamma)}} - \gamma (\beta_2^*)^{\frac{1}{1 - (\alpha + \gamma)}} \right]^{\frac{1 - (\alpha + \gamma)}{\alpha}}, \tag{19}$$

if $S_2^* > 0$. Otherwise, $\widehat{\theta} = +\infty$.

Condition (19) illustrates Proposition 3. When the cost of administrative tasks for J is lower than the threshold, the optimal contract is such that J performs the two tasks (contract (C1)) and, when the cost of administrative tasks for J is larger than this threshold, the optimal contract is such that J does not provide any administrative effort (contract (C2)).

The threshold, characterized by the right-hand side in condition (19), does not depend on the cost functions because the judicial production function has a Cobb-Douglas form. It depends only on the administrative tasks (α) and jurisdictional tasks (γ) output elasticities.

The relationship between the threshold and the two elasticities is not straightforward. In order to obtain further insights, Table 1 provides numerical computations of the threshold as a function of the two elasticities. The first row in Table 1 presents the value of the judicial output elasticity of jurisdictional effort γ (we let the value of this parameter vary between 0.1 and 0.9). The first row indicates the value of the judicial output elasticity of administrative effort α (we make the value of this parameter vary between 0.01 and 0.9).

This numerical example deserves at least three comments. First, we observe that the threshold increases when γ increases. In other words, when the output elasticity of the jurisdictional effort increases, the likelihood that the optimal contract is such that the judge performs both kinds of tasks (contract (C1)) increases. Second, and somewhat surprisingly, we observe that the threshold can decrease and then increase when α increases: for instance, if $\gamma = 0.7$, then $\hat{\theta}/\delta_r = 2082.10$

	γ	0.1	0.3	0.5	0.7	0.9
α						
0.01		3.12	9.09	50.23	4485.20	$+\infty$
0.02		2.41	5.91	25.97	2082.10	$+\infty$
0.03		2.11	4.67	18.00	1689.10	$+\infty$
0.05		1.82	3.56	11.67	3181.60	$+\infty$
0.1		1.56	2.60	6.90	$+\infty$	_
0.3		1.37	1.90	3.76	_	_
0.4		1.39	1.91	3.52	_	_
0.5		1.47	2.21	_	_	_
0.7		$+\infty$	_	_	_	_
0.9		_	_	_	_	_

Table 1: Threshold $\hat{\theta}/\delta_r$ as a function of α and γ

when $\alpha = 0.02$, $\theta/\delta_r = 1689.10$ when $\alpha = 0.03$ and $\theta/\delta_r = 3181.60$ when $\alpha = 0.05$. In other words, when the output elasticity of administrative effort increases, the likelihood that the optimal contract is such that J performs both tasks (contract (C1)) may first decrease and then increase. Third, when γ is sufficiently large, the threshold goes to infinity. In other words, when the output elasticity of jurisdictional task is sufficiently large, the optimal contract is such that J exerts effort on the two tasks (contract (C1)).

How big is the distortion of the optimal efforts? We now can evaluate how much the effort levels differ between the first-best situation, and under the two contracts (C1) and (C2). Let us compute the rate of decrease in the jurisdictional effort and the total administrative effort when switching from the first-best situation to contract (C1). These rates are given by:

$$\frac{e^{FB} - e_1^*}{e^{FB}} = 1 - \left(\frac{\delta_r}{\delta_a}\right)^{\frac{\alpha}{1 - (\alpha + \gamma)}} \text{ and } \frac{A^{FB} - A_1^*}{A^{FB}} = 1 - \left(\frac{\delta_r}{\delta_a}\right)^{\frac{1 - \gamma}{1 - (\alpha + \gamma)}}.$$
 (20)

The two rates increase when the ratio of the cost of administrative effort δ_r/δ_a (which is smaller than 1) decreases, when α increases and when γ increases. In other words, in the case of contract (C1), both the jurisdictional effort and the administrative effort move away from the first-best levels when J's cost of performing administrative tasks increases relatively to G's cost of performing administrative tasks and when the judicial output becomes more elastic with respect to administrative or jurisdictional efforts. Note that the fact that the degree of inefficiency in the effort level depends positively on the ratio of the unit costs of administrative tasks is intuitive. Indeed, the inefficiency of contract (C1) is due to the fact that, unlike in the first-best situation, J provides administrative efforts, and then, the higher the cost of performing these tasks for J compared to G, the higher the degree of inefficiency of contract (C1).

We now do the same kind of analysis for contract (C2) which is such that J does not perform any administrative tasks. When switching from the first-best situation to contract (C2), the rates of decrease of the jurisdictional effort and of the total administrative effort are given by:

$$\frac{e^{FB} - e_2^*}{e^{FB}} = 1 - (\beta_2^*)^{\frac{1}{1 - (\alpha + \gamma)}} \text{ and } \frac{A^{FB} - A_2^*}{A^{FB}} = 1 - (1 - \beta_2^*)^{\frac{1 - \gamma}{1 - (\alpha + \gamma)}}, \tag{21}$$

where β_2^* is characterized by condition (16).

Notice that, when the judicial output function has a Cobb-Douglas form, these rates do not depend on the cost parameters but only on the output elasticities of the administrative and jurisdictional efforts, α and γ . In particular, the rate of decrease of the jurisdictional effort increases unambiguously as α increases, and the rate of decrease of the administrative effort increases unambiguously as γ increases. However, the effect of an increase in γ on the rate of decrease of the jurisdictional effort, and the effect of an increase in α on the rate of decrease of the administrative effort are ambiguous because β_2^* increases when γ increases and decreases when α increases, and the weight $\frac{1}{1-(\alpha+\gamma)}$ increases when γ or α increase. Overall, in the case of contract (C2), we can only conclude that the jurisdictional effort moves away from the first-best level if the judicial output is more sensitive to administrative efforts, while the administrative effort moves away from the first best level if the judicial output is more sensitive to jurisdictional efforts.

How large is the loss of total surplus? In order to evaluate the extent of the loss in total surplus induced by the two contracts (C1) and (C2), we compute the rate of decrease of the total surplus. When switching from the first best situation to contract (C1), we have:

$$\frac{S^{FB} - S_1^*}{S^{FB}} = 1 - \left(\frac{\delta_r}{\delta_a}\right)^{\frac{\alpha}{1 - (\alpha + \gamma)}}.$$
 (22)

The degree of inefficiency, measured by the rate of decrease of the total surplus, is higher if J's cost of administrative effort increases compared to G's, and if the judicial production function becomes more elastic with respect to administrative efforts or jurisdictional efforts.

When switching from the first-best situation to contract (C2), we have:

$$\frac{S^{FB} - S_2^*}{S^{FB}} = 1 - \frac{\left[(1 - \beta_2^*)^{\alpha} (\beta_2^*)^{\gamma} \right]^{\frac{1}{1 - (\alpha + \gamma)}} - \alpha (1 - \beta_2^*)^{\frac{1}{1 - (\alpha + \gamma)}} - \gamma (\beta_2^*)^{\frac{1}{1 - (\alpha + \gamma)}}}{1 - (\alpha + \gamma)}.$$
 (23)

The relationship between this rate of decrease and the two elasticities is not straightforward. In order to obtain some insights into this relationship, we provide numerical computations of this decrease rate as a function of the two elasticities in Table 2.

	γ	0.1	0.3	0.5	0.7	0.9
α						
0.01		1	3	8	25	247
0.02		2	5	13	42	387
0.03		3	7	17	56	471
0.05		3	9	24	80	475
0.1		5	15	38	124	_
0.3		15	38	86	_	_
0.5		38	86	_	_	_
0.7		124	_	_	_	_
0.9		_	_	_	_	_

Table 2: Rate of decrease of the total surplus (%) as a function of α and γ (contract (C2)).

In Table 2, we observe that the rate of decrease of the total surplus increases when the output

elasticities of administrative and jurisdictional tasks increase.

5 Extensions

In this section, we relax several of our model assumptions and study whether this leads to the conclusion that beyond the minimal level of administrative tasks \underline{r} , both the government and the judge provide additional administrative efforts. First, we introduce a time constraint for the judge and then consider that the judge's and the government's administrative efforts are imperfect substitutes.

5.1 Judge's time constraint

Assume that J faces a time constraint $a + e \leq T$. Would this lead J and G to invest effort in administrative tasks? To answer this question, let us denote $\mu_T \geq 0$ the Lagrangian multiplier associated with the time constraint $a + e \leq T$. The necessary conditions for J and G to choose respectively a > 0 and r > 0 can be written as follows:

$$f_A - \delta_a + \lambda_a (a\beta f_{AA}) + \lambda_e e\beta f_{Ae} + \lambda_r r(1 - \beta) f_{AA} - \mu_T = 0.$$
 (24)

and,

$$f_A - \delta_r + \lambda_a a \beta f_{AA} + \lambda_e e \beta f_{Ae} + \lambda_r r (1 - \beta) f_{AA} = 0$$
 (25)

Combining these two conditions, we obtain $\mu_T = \delta_r - \delta_a < 0$, which is a contradiction.

Hence, if G chooses a contract in which J provides administrative tasks (a > 0), then G will not perform more administrative tasks than the minimum level (r = 0). If G chooses a contract in which G performs more administrative tasks than the minimum level (r > 0) then J will not perform any administrative tasks (a = 0).

5.2 Administrative task: imperfect substitutability and convex costs

We now assume that the efforts of G and J on administrative tasks are imperfectly substitutable. Let us assume that A = a + h(r) with h' > 0 and $h'' \le 0$. Note that h' represents the marginal productivity of G's effort on administrative tasks whereas the marginal productivity of J's effort on administrative tasks is 1. The following two necessary conditions have to hold for both a and r to be strictly positive:

$$f_A - \delta_a + \lambda_a a \beta f_{AA} + \lambda_e e \beta f_{Ae} + \lambda_r r (1 - \beta) f_{AA} = 0. \tag{26}$$

and,

$$h'f_A - \delta_r + \lambda_a a\beta f_{AA} + \lambda_e e\beta f_{Ae} + \lambda_r r(1 - \beta) f_{AA} = 0$$
(27)

These conditions hold only if $(1 - h')f_A = \delta_a - \delta_r$ with $\delta_a - \delta_r > 0$. Hence, it is necessary to have h' < 1. In this case, it is possible that G and J share administrative tasks, and G exerts an effort larger than the minimum, i.e. a > 0 and r > 0. However, it is necessary that J's effort in the administrative task is more productive than G's effort (1 > h'), which is a debatable assumption.

Notice that the assumption of imperfect substitutability with A = a + h(r) is equivalent to assuming that G's administrative effort cost is convex. Let R = h(r) denote the effective

administrative effort of G. Then the cost of exerting an effective effort level R is $\delta_r r = \delta_r h^{-1}(R)$, where h^{-1} is the inverse function of h. Since h is concave, then h^{-1} is convex. In other words, the cost of exerting an effective effort R is convex.

6 Discussions

In this section, we discuss possible solutions to the problem at hand and several testable implications of our analysis.

6.1 Remedies

We discuss two possible solutions and the extent to which they are implementable. First, we discuss how the problem could be resolved if the government were able to commit to providing a specific level of administrative effort, and why it may prefer not to do this. We then discuss the implication of a policy that would ban judges from carrying out administrative tasks.

6.1.1 The government's commitment issue

Our model is based on the assumption that the government does not commit to providing a sufficiently high pre-specified level of administrative tasks, \underline{r} , prior to setting the incentive scheme. In this sense, the contract we have analyzed is an *incomplete contract*. Obviously, if the government could commit to setting the first-best level of administrative tasks – i.e. $\underline{r} = r^{FB}$ – then it could implement the first-best while choosing to provide high-powered incentives to the judge ($\beta = 1$). There are at least two reasons why we believe that it is reasonable to assume that the government cannot commit to setting the first-best level of administrative tasks. First, governments generally lack commitment power; second, governments may lack the information necessary to estimate the first-best level (at least ex-ante). We discuss these reasons in detail below.

Governments are comprised of elected politicians and members who are elected or appointed for relatively short periods of time. Therefore, the composition of governments changes frequently, and the promises and decisions made by one government are not necessarily honored by the government or by the succeeding ones. One way for the government to commit to providing the first-best level of administrative effort (assuming that this level can be known), is by being constrained to obey a rule that does not allow it to violate past decisions (North and Weingast, 1989; Acemoglu et al., 2010). In reality, such institutional bounds are rarely observed, and also are not always beneficial. Indeed, in some situations, it has been shown that incomplete contracts can outperform complete contracts (Bernheim and Whinston, 1998; Battaglini and Harstad, 2016; Crocker and Masten, 1991; Crocker and Reynolds, 1993). For instance, Crocker and Masten (1991) argue that in the context of an uncertain and complex environment, it can be more efficient if parties do not commit ex ante to specific decisions because it allows them ex post to adapt to changing economic circumstances. In our context, we assume that the government can commit to providing a minimal level of administrative effort, i.e. it can choose r before choosing the incentive scheme. At the time that government chooses this minimal level, there will likely be some uncertainty about the future marginal opportunity cost of judicial funds, i.e. there is likely to be some uncertainty about parameter δ_r . Indeed, the opportunity cost of judicial funds is likely to vary depending on budgetary pressures and financial necessity. Hence, in our setting,

the government may prefer not to commit ex ante to providing a specific level of administrative tasks in order to benefit from flexibility to choose the level of its administrative effort after the uncertainty is resolved.

6.1.2 Banning the judge from performing administrative tasks

It might be thought that one solution to the problem would be (when feasible) to ban the judge from performing administrative tasks. However, in our setting, this policy would not increase the total surplus. Indeed, banning the judge from carrying out administrative tasks implies that the government has to consider an additional constraint, a=0. As a consequence, the total surplus (which the government maximizes) cannot increase. Indeed, with a ban, when it would be optimal to implement the non-specialization contract with high-powered incentives, i.e. when the unit cost of administrative task of the judge is sufficiently low (see Proposition 3), the government would be forced to choose the specialization contract (C2), and then would receive a lower surplus than in the case of the non-specialization contract (C1). More generally, a mandatory restriction on the judge's administrative effort cannot lead to an increase in the total surplus.

6.2 Predictions and testable implications

In this section, we provide several implications of our model. First, we discuss some implications related to the assessment of the effectiveness of the judiciary. Then we deliver a testable implication of our model in relation to skill-biased technical change in courts. We also provide a testable implication regarding the link between economic shocks, judge incentive schemes and the induced allocation of tasks.

6.2.1 Assessment and errors

We have assumed that the government, at the time it chooses the optimal contract, perfectly knows the fundamentals of the model. However, note that a minimal error in assessing the cost of the administrative tasks for instance, could have a large negative impact. Indeed, since the nature of the optimal contract (C1 or C2) depends on whether the ratio of the unit cost of the government's administrative effort compared to the judge's, is below or above a specific threshold, then a small error could lead government to choose contract (C1) while (C2) is optimal, or to choose contract (C1) while contract (C2) is optimal. For instance, if the cost of the administrative effort of the government is overestimated, then the government might choose contract (C1) while contract (C2) is the optimal one, resulting in the judge performing administrative tasks when it would be optimal for the judge not to do so. In contrast, if the cost of the government's administrative effort is underestimated, then the government could choose contract (C2) while contract (C1) would be optimal and the judge then would not carry out any administrative tasks when it would be optimal for her to do so.

The analysis in the main part of the paper and the above remarks have an important implication for the measurement of the effectiveness of the judiciary. They highlight that judiciaries where judges do not perform administrative tasks are not necessarily efficient since this kind of arrangement could correspond to a second-best situation (when judges are able to perform administrative tasks at low cost), or even to a third-best situation (e.g. when the cost of judges

performing administrative tasks has been overestimated).

6.2.2 Skill-biased technical change

Our model predicts that if the relative cost of judges performing administrative tasks becomes sufficiently low, the government should provide strong incentives for them to perform both jurisdictional and administrative tasks instead of just jurisdictional tasks. The relative cost of administrative tasks performed by judges compared to administrative staff may be affected by new technologies. Moreover, new technologies such as computers are likely to be biased in favor of the most skilled (Autor et al., 1998), the judge in our case.

Therefore, our analysis provides predictions about the link between skill-biased technical change, the distribution of earnings in the organization, and the resulting allocation of tasks. For instance, assuming that computerization of courts enables judges to perform some administrative tasks at a lower cost compared to their being conducted by administrative staff, our model implies that computerization of courts increases the likelihood that judges perform administrative tasks and face higher incentives.

6.2.3 Economic shocks

Any shock that affects the government's use of scarce public resources can lead to a change in the (opportunity) cost of the administrative tasks for the government. For instance, in the context of economic and budgetary crises, allocating public resources implies difficult trade-offs.

Assume, for example, that the initial optimal contract is such that the incentives are shared and judges perform jurisdictional tasks while the government carries out administrative tasks. A negative economic shock could lead to a large increase in the opportunity cost of administrative tasks being performed by government, and the optimal contract may switch from a situation with shared incentives to a situation with high-powered incentives.

Assuming that the government is able to adapt judges' incentive schemes sufficiently rapidly, our model delivers clear predictions about how judges' incentives should vary following an economic shock, and how this would affect the allocation of tasks: judges are likely to face higher incentives and to perform more administrative tasks following a negative economic shock but to face lower incentives and perform fewer administrative tasks following a positive economic shock.

7 Conclusion

In this paper, we provide a double-sided moral hazard model of the production of justice. The judicial output depends on jurisdictional tasks that can only be provided by a judge, and on administrative tasks that can be provided by either the government and/or by the judge. However, the judge performs the administrative task at a higher unit cost. In this setting, the first-best situation is that the judge specializes in jurisdictional tasks.

When the efforts of the government and the judge are not contractible, there is a double-sided moral hazard problem. We find that the optimal contract is either such that the judge faces very high-powered incentives and provides jurisdictional and administrative tasks, or the incentives are shared between the judge and the government with the result that the judge specializes in jurisdictional tasks and the government provides administrative tasks.

When deciding between these two forms of contract, the government faces a trade-off between two different sources of inefficiency. In the former case, there is a misallocation of the administrative tasks, while in the latter case the government and the judge do not reap the full returns from their efforts and their effort levels are distorted compared to the first-best situation. We show that the form of the optimal contract depends on the difference between the government's and the judge's unit cost of performing the administrative task. When the differential is large, the second-best optimal contract is the one where incentives are shared and the judge does not perform any administrative tasks. When the differential is small, the second-best optimal contract is the one where the judge faces high-powered incentives and performs both jurisdictional and administrative tasks.

Our analysis highlights that judicial systems in which judges do not perform administrative tasks are not necessarily (more) efficient. Indeed, if the (double) moral hazard inefficiency is less severe than the misallocation inefficiency, the second-best optimal situation is such that judges face low-powered incentives and specialize in jurisdictional tasks. This is the case especially if judges are able to perform administrative tasks at relatively low cost.

Our analysis also provides a range of testable implications. For instance, our results suggest that skill-biased technical change or negative economic shocks increase the likelihood that judges face very high-powered incentives and perform administrative tasks.

Lack of commitment power on the part of the government is central to our analysis. Increasing this power would have benefits where the government is able to assess the relevant level of administrative tasks ex ante. However, if there is uncertainty about this relevant level, it would be better for the government not to commit and to benefit from some flexibility.

We believe that there is no straightforward solution to eliminating these two inefficiencies. Our analysis suggests that simple solutions such as banning judges from doing administrative tasks (assuming this is feasible), are likely to have a negative impact. Banning judges from doing administrative tasks would constrain the government from choosing the contract where incentives are shared even when this is not optimal. In other words, it would force the government to choose the (double) moral hazard inefficiency even when the misallocation inefficiency is less severe.

Appendix: Additional Proofs

Proof of Proposition 2: We consider different situations.

Case I: a > 0 and r > 0:

We must have $\mu_a = \mu_r = 0$ and then (NC_a) and (NC_r) imply that $\delta_a = \delta_r$, which is a contradiction.

Case II: a > 0, e > 0 and r = 0:

We must have $\mu_a = \mu_e = 0$ and $\beta f_A = \delta_a$, $\beta f_e = c'$ and $(1 - \beta)f_A - \delta_r \leq 0$. Plugging these conditions into (NC_a) , (NC_e) , (NC_r) and (NC_β) we find:

$$(1-\beta)f_A + \beta \lambda_a a(f_{AA} - \frac{f_A}{f_e} f_{Ae}) = 0$$
(28)

$$(1-\beta)f_e + \lambda_a a \left[\beta (f_{Ae} - \frac{f_A}{f_e} f_{ee}) + \frac{f_A}{f_e} c'' \right] = 0$$
 (29)

$$f_A - \delta_r + \lambda_a a \beta f_{AA} + \lambda_e e \beta f_{Ae} + \lambda_r \left[(1 - \beta) f_A - \delta_r \right] + \mu_r = 0$$
(30)

$$\lambda_e e = -\lambda_a a f_A / f_e \tag{31}$$

If $f_A f_{Ae} - f_e f_{AA} = 0$, then condition (28) implies that $\beta = 1$. If $f_A f_{Ae} - f_e f_{AA} \neq 0$, conditions (28) and (29) imply that $\beta > 0$ and:

$$\lambda_a a = \frac{1 - \beta}{\beta} \frac{f_A f_e}{f_A f_{Ae} - f_e f_{AA}},\tag{32}$$

$$(1-\beta)(f_e)^2 + \frac{1-\beta}{\beta} \frac{f_A f_e}{f_A f_{Ae} - f_e f_{AA}} \left[\beta (f_e f_{Ae} - f_A f_{ee}) + f_A c'' \right] = 0$$
 (33)

(34)

One can easily check that condition (33) implies that $\beta = 1$ and then $\lambda_a = 0$, $f_A = \delta_a$, and $f_e = c'$. Notice that $(1 - \beta)f_A - \delta_r = -\delta_r \le 0$.

Case III: a = 0, e > 0 and r > 0:

We must have $\mu_e = \mu_r = 0$, $\beta f_e = c'$, $(1 - \beta)f_A = \delta_r$ and $\beta f_A - \delta_a \leq 0$. One can easily check that one must have $0 < \beta < 1$. Moreover, plugging these conditions into (NC_a) , (NC_e) , (NC_r) and (NC_{β}) we find:

$$f_A - \delta_a + \lambda_a (\beta f_A - \delta_a) + \lambda_e e \beta f_{Ae} + \lambda_r r (1 - \beta) f_{AA} + \mu_a = 0, \tag{35}$$

$$\lambda_r r \left(f_{Ae} - \frac{f_A}{f_e} c'' + \beta \left(\frac{f_A}{f_e} f_{ee} - f_{Ae} \right) \right) = -(1 - \beta) f_e, \tag{36}$$

$$\lambda_r r \left(f_{AA} + \beta \left(\frac{f_A}{f_e} f_{Ae} - f_{AA} \right) \right) = -\beta f_A, \tag{37}$$

$$\lambda_e e f_e = \lambda_r r f_A \tag{38}$$

We then must have $\lambda_r \neq 0$ and $f_{AA} + \beta \left(\frac{f_A}{f_e} f_{Ae} - f_{AA}\right) \neq 0$. We can then easily show that

$$\lambda_r = -\frac{f_A - \delta_r}{f_{AA} + \beta \left(\frac{f_A}{f_e} f_{Ae} - f_{AA}\right)}$$
 and:

$$[(f_A)^2 f_{ee} - (f_e)^2 f_{AA}] \beta^2 + [2(f_e)^2 f_{AA} - (f_A)^2 c''] \beta - (f_e)^2 f_{AA} = 0$$
(39)

If $(f_A)^2 f_{ee} - (f_e)^2 f_{AA} = 0$ then $\beta = \frac{-(f_e)^2 f_{AA}}{(f_A)^2 c'' - 2(f_e)^2 f_{AA}}$. Otherwise,

$$\beta = \frac{(f_A)^2 c'' - 2(f_e)^2 f_{AA} - \sqrt{D}}{2\left[(f_A)^2 f_{ee} - (f_e)^2 f_{AA} \right]},\tag{40}$$

where

$$D = \left[2(f_e)^2 f_{AA} - (f_A)^2 c''\right]^2 + 4(f_e)^2 f_{AA} \left[(f_A)^2 f_{ee} - (f_e)^2 f_{AA}\right]$$
(41)

$$= (f_A)^2 \left[(f_A)^2 (c'')^2 - 4(f_e)^2 f_{AA}(c'' - f_{ee}) \right] > 0.$$
(42)

Notice that the second root of (39) cannot be a solution since $\frac{(f_A)^2c''-2(f_e)^2f_{AA}+\sqrt{D}}{2[(f_A)^2f_{ee}-(f_e)^2f_{AA}]} > 1$. Also notice that the right-hand side in (40) belongs to (0,1).

Case IV: a = 0, e > 0 and r = 0:

We must have $\mu_e = 0$, $\beta f_e = c'$, $(1 - \beta)f_A - \delta_r \le 0$ and $\beta f_A - \delta_a \le 0$. Plugging these conditions into (NC_a) , (NC_e) , (NC_r) and (NC_{β}) we find:

$$f_A - \delta_a + \lambda_a \left(\beta f_A - \delta_a\right) + \mu_a = 0 \tag{43}$$

$$f_e - c' = 0.$$
 (44)

$$f_A - \delta_r + \lambda_r \left[(1 - \beta) f_A - \delta_r \right] + \mu_r = 0. \tag{45}$$

$$\lambda_e = 0. (46)$$

We then must have $\beta = 1$. It remains to check that $\psi(\underline{r}) \equiv f_A(\underline{r}, e) - \delta_a \leq 0$, which holds because of Assumption 1.

Proof of Corollary 1: In the case of contract (C1), we have $f_A = \delta_a$ and $f_e = c'$. Differentiating with respect to δ_a , we find $\frac{\partial e^*}{\partial \delta_a} = \frac{f_{eA}}{c'' - f_{ee}} \frac{\partial A^*}{\partial \delta_a}$ and $\frac{\partial A^*}{\partial \delta_a} = 1/\left(c'' f_{AA} - f_{AA} f_{ee} + (f_{eA})^2\right) < 0$. Since $\delta_a > \delta_r$, we have $e^* < e^{FB}$ and $A^* < A^{FB}$.

Now let us define two parameters, $\theta_1 > 1$ and $\theta_2 > 1$ such that $\theta_1 f_A = \delta_a$ and $\theta_2 f_e = c'$. The effort levels in case (C2) are characterized by these two conditions when $\theta_2 = \beta$ and $\theta_1 = 1 - \beta$. The first-best effort levels are characterized by the two conditions when $\theta_2 = 1$ and $\theta_1 = 1$.

Differentiating the two conditions with respect to θ_1 and θ_2 , we find:

$$\frac{\partial A}{\partial \theta_1} = -\frac{(c'' - \theta_2 f_{ee}) f_A}{\theta_1 c'' f_{AA} - \theta_1 \theta_2 (f_{AA} f_{ee} - (f_{Ae})^2)} > 0,$$

$$\frac{\partial e}{\partial \theta_1} = -\frac{\theta_2 f_A f_{eA}}{\theta_1 c'' f_{AA} - \theta_1 \theta_2 (f_{AA} f_{ee} - (f_{Ae})^2)} \ge 0,$$

$$\frac{\partial A}{\partial \theta_2} = \frac{f_{Ae} f_e}{\theta_2 (f_{AA} f_{ee} - (f_{Ae})^2)} \ge 0,$$
(48)

$$\frac{\partial e}{\partial \theta_1} = -\frac{\theta_2 f_A f_{eA}}{\theta_1 c'' f_{AA} - \theta_1 \theta_2 (f_{AA} f_{ee} - (f_{Ae})^2)} \ge 0,\tag{48}$$

$$\frac{\partial A}{\partial \theta_2} = \frac{f_{Ae}f_e}{\theta_2(f_{AA}f_{ee} - (f_{Ae})^2)} \ge 0,\tag{49}$$

$$\frac{\partial e}{\partial \theta_2} = -\frac{f_{AA}f_e}{\theta_2(f_{AA}f_{ee} - (f_{Ae})^2)} > 0. \tag{50}$$

This is sufficient to complete the proof.

Proof of Proposition 3: Let a_1^*, e_1^* denote the optimal levels of effort for contract (C1) and r_2^*, e_2^* denote the optimal levels of effort for contract (C2). Contract (C1) is preferred to contract (C2) if and only if:

$$\Delta = f(A_1^*, e_1^*) - \delta_a a_1^* - c(e_1^*) - [f(A_2^*, e_2^*) - \delta_r r_2^* - c(e_2^*)] \ge 0, \tag{51}$$

where $A_1^* = a_1^* + \underline{r}$ and $A_2^* = r_2^* + \underline{r}$. (A_1^*, e_1^*) is characterized by $f_A(A_1^*, e_1^*) = \delta_a$ and $f_e(A_1^*, e_1^*) = c'(e_1^*)$ and (A_2^*, e_2^*) is characterized by $(1 - \beta^*) f_A(A_2^*, e_2^*) = \delta_r$, $\beta^* f_e(A_2^*, e_2^*) = c'(e_2^*)$ and β^* is characterized as in condition (40). We have $\frac{\partial \Delta}{\partial \delta_a} = -a_1^* < 0$. Moreover, $\lim_{\delta_a \to \delta_r} \Delta = f(A^{FB}, e^{FB}) - \delta_r A^{FB} - c(e^{FB}) - [f(A_2^*, e_2^*) - \delta_r r_2^* - c(e_2^*)] > 0$.

References

- Acemoglu, D., M. Golosov, and A. Tsyvinsky (2010). Dynamic mirrlees taxation under political economy constraints. *Review of Economic Studies* 77(3), 841–881.
- Autor, D. H., L. F. Katz, and A. B. Krueger (1998). Computing inequality: Have computers changed the labor market? *Quarterly Journal of Economics* 113, 1169–1213.
- Bagues, M. and B. Esteve-Volart (2010). Performance pay and judicial production: Evidence from spain. *Unpublished manuscript*.
- Battaglini, M. and B. Harstad (2016). Participation and duration of environmental agreements. *Journal of Political Economy* 124(1), 160–204.
- Beenstock, M. and Y. Haitovsky (2004). Does the appointment of judges increase the output of the judiciary? *International Review of Law and Economics* 24(3), 351 369.
- Bernheim, B. D. and M. Whinston (1998). Incomplete contracts and strategic ambiguity. *American Economic Review* 88(4), 902–32.
- Bhattacharyya, S. and F. Lafontaine (1995). Double-sided moral hazard and the nature of share contracts. RAND Journal of Economics 26(4), 761–781.
- Burki, S. J. and G. Perry (1998). Beyond the Washington consensus: institutions matter. World Bank Publications.
- Buscaglia, E. and M. Dakolias (1999). Comparative international study of court performance indicators: a descriptive and analytical account. Technical report, Washington, DC: World Bank.
- Buscaglia, E. and T. Ulen (1997). A quantitative assessment of the efficiency of the judicial sector in latin america. *International Review of Law and Economics* 17(2), 275–291.
- Choi, S. J., G. M. Gulati, and E. A. Posner (2010). Professionals or politicians: The uncertain empirical case for an elected rather than appointed judiciary. *Journal of Law, Economics, and Organization* 26(2), 290–336.
- Chopard, B., E. Fain, and L. Roussey (2018). Does the appeals process reduce the occurrence of legal errors? *Review of Law & Economics* (Forthcoming).
- Cooter, R. D. (1983). The objectives of private and public judges. Public Choice 41(1), 107–132.

- Crocker, K. J. and S. E. Masten (1991). Pretia ex machina? prices and process in long-term contracts. The Journal of Law and Economics 34(1), 69–99.
- Crocker, K. J. and K. J. Reynolds (1993). The efficiency of incomplete contracts: an empirical analysis of air force engine procurement. *The RAND journal of economics*, 126–146.
- Dakolias, M. (1999). Court performance around the world: a comparative perspective, Volume 23. World Bank Publications.
- Deffains, B. and L. Roussey (2011). The multiplier effect of public expenditure on justice: The case of rental litigation. *Review of Law & Economics* 7(1), 243–263.
- Dewatripont, M. and J. Tirole (1999). Advocates. Journal of Political Economy 107(1), 1–39.
- Dimitrova-Grajzl, V., P. Grajzl, J. Sustersic, and K. Zajc (2012). Court output, judicial staffing, and the demand for court services: Evidence from slovenian courts of first instance. *International Review of Law and Economics* 32(1), 19–29.
- Dimitrova-Grajzl, V., P. Grajzl, and K. Zajc (2014). Understanding modes of civil case disposition: Evidence from slovenian courts. *Journal of Comparative Economics* 42(4), 924 939.
- Djankov, S., R. La Porta, F. Lopez-de Silanes, and A. Shleifer (2003). Courts. Quarterly Journal of Economics 118(2), 453-517.
- Eswaran, M. and A. Kotwal (1985). A theory of contractual structure in agriculture. *American Economic Review* 75(3), 352–367.
- Flanders, S. (1977). Case management and court management in United States district courts, Volume 77. Federal Judicial Center.
- French, H. (2009). Executive toys: Judges and non-judicial functions. *Journal of Judicial Administration* 19, 5–13.
- Garoupa, N. M. and T. Ginsburg (2010). Reputation, information and the organization of the judiciary. Journal of Comparative Law 4, 226–254.
- Gomes, A.O., G. T. and L. Akutsu (2016). The relationship between judicial staff and court performance: Evidence from brazilian state courts. *International Journal for Court Administration* 1(8), 12–19.
- Grajzl, P. and S. Silwal (2016). Courts in a resource-starved developing economy: Case disposition and the quantity-quality tradeoff in post-conflict nepal. Working Paper.
- Holmstrom, B. (1982). Moral hazard in teams. The Bell Journal of Economics 13(2), 324-340.
- La Porta, R., F. Lopez-de Silanes, A. Shleifer, and R. W. Vishny (1997). Legal determinants of external finance. *Journal of Finance* 52(3), 1131–1150.
- La Porta, R., F. Lopez-de Silanes, A. Shleifer, and R. W. Vishny (1998). Law and finance. *Journal of Political Economy* 106(6), 1113–1155.
- Lal, R. (1990). Improving channel coordination through franchising. Marketing Science 9(4), 299–318.
- Landes, W. M. and R. A. Posner (1979). Adjudication as a private good. *The Journal of Legal Studies* 8(2), 235–284.
- Legros, P. and H. Matsushima (1991). Efficiency in partnerships. *Journal of Economic Theory* 55(2), 296 322.

- Legros, P. and S. A. Matthews (1993). Efficient and Nearly-Efficient Partnerships. *Review of Economic Studies* 60(3), 599–611.
- Mathewson, G. F. and R. A. Winter (1985). The economics of franchise contracts. *Journal of Law & Economics 28*(3), 503–526.
- Mendel, T. (2015). Efficiency run amok: Challenging the authority of magistrate judges to hear and accept felony guilty pleas. *Vanderbilt Law Review* 68(6), 1795–1832.
- Mitsopoulos, M. and T. Pelagidis (2007). Does staffing affect the time to dispose cases in greek courts? *International Review of Law and Economics* 27(2), 219–244.
- New York State Committee to Review Audio-Visual Coverage of Court Proceedings (1997). An Open Courtroom: Cameras in New York Courts. New York: Fordham University Press.
- North, D. C. (1990). *Institutions, institutional change and economic performance*. Cambridge university press.
- North, D. C. and B. R. Weingast (1989). Constitutions and commitment: The evolution of institutions governing public choice in seventeenth-century england. *Journal of Economic History* 49(4), 803–832.
- Peyrache, A. and A. Zago (2016). Large courts, small justice!: The inefficiency and the optimal structure of the italian justice sector. *Omega 64* (Supplement C), 42 56.
- Posner, R. A. (1993). What do judges and justices maximize? (the same thing everybody else does). Supreme Court Economic Review 3, 1–41.
- Rosales-López, V. (2008). Economics of court performance: an empirical analysis. European Journal of Law and Economics 25(3), 231–251.
- Roussey, L. (2016). Enjeux théoriques et opérationnels de l'analyse économique du système judiciaire. Économie et institutions (24), 5–40.
- Roussey, L. and B. Deffains (2012). Trust in judicial institutions: An empirical approach. *Journal of Institutional Economics* 8(3), 351–369.
- Schneider, M. R. (2005). Judicial career incentives and court performance: An empirical study of the german labour courts of appeal. *European Journal of Law and Economics* 20(2), 127–144.
- Shapiro, S. A. and R. E. Levy (1995). Judicial incentives and indeterminacy in substantive review of administrative decisions. *Duke Law Journal* 44(4), 1051–1080.
- Sheldon, C. H. (1981). Law clerking with a state supreme court: Views from the perspective of the personal assistants to the judges. *The Justice System Journal* 6(3), 346–371.
- Smith, C. E. and H. Feldman (2001). Burdens of the bench: state supreme courts' non-judicial tasks. *Judicature* 84(6), 304–309.
- Stiglitz, J. E. (1974). Incentives and risk sharing in sharecropping. Review of Economic Studies 41(2), 219–255.
- Taha, A. E. (2004). Publish or paris? evidence of how judges allocate their time. American Law and Economics Review 6(1), 1–27.

CEE-M Working Papers1 - 2018

WP 2018 - 01: Ilaria Brunetti., Mabbel Tidball, & Denis Couvet

« Relationship Between Biodiversity and Agricultural Production »

WP 2018 - 02: Phillippe Le Coent, Raphaële Préget & Sophie Thoyer

« Do farmers follow the herd? The influence of social norms in the

participation to agri-environmental schemes.»

WP 2018 - 03: Ludivine Roussey & Raphael Soubeyran

« Overburdened judges »